Author’s response to reviews

Title: Treatment of thulium laser for upper tract urothelial carcinoma via ureteroscopy: a retrospective monocentric study

Authors:

Jin Wen (wjjxmc@163.com)
Zhi Ji (jzgjxmc@163.com)
Han Li (lhzjxmc@163.com)

Version: 4 Date: 05 Feb 2018

Author’s response to reviews:

Dear Reviewers:

Thanks very much for your advice. According to your suggestions, we revised the manuscript. The point-by-point response is listed below.

1. All patients provided written informed consent, and this is stated in the section of “Ethics approval and consent to participate”.

2. A Conclusion heading is added to the manuscript.

3. The email addresses and affiliated institutions for all authors are provided in the section of “Authors details”.

4. Authors’ names are represented using their full initials in the section of “Authors’ contributions” as required.

5. A final and clean version will be uploaded.
Sincerely

Pro Jin Wen

wpumch@163.com

Pro Zhi G Ji

jzgjxmc@163.com